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ABSTRACT:A system combining nonthermal plasma (NTP) and catalysis using Mn Co/HZSM-5 catalyst has been
developed to promote the oxidation of acetaldehyde and reduce the production of ozone. The introduction of catalyst improved
the oxidation of acetaldehyde and inhibited the generation of O5 in comparison with the NTP alone system. The mixed MnCo/
HZSM-5 catalyst demonstrated the highest catalytic activity among the various prepared catalysts, and this was ascribed to its
crystalline structure, redox property, and adsorbed surface oxygen of the catalyst. The removal e ciency of acetaldehyde and
CO, vield were positively associated with the speci ¢ energy density (SED), and they were inversely related to the gas ow rate.
Nevertheless, the energy yield was in negative correlation with the SED and in positive association with the gas ow rate.
Moreover, the impact of di erent catalysts on the acceleration of acetaldehyde degradation in the hybrid system was discussed

by developing a simple kinetic model.

1. INTRODUCTION

Volatile organic compounds (VOCs) which are released from
multiple transportation vehicles and industries form a
dominant source of air pollution." Due to their malodorous,
toxic, mutagenic, and carcmogenlc nature, the removal of
VOCs has gained public concern.? VOCs are also associated
with the formation of photochemical smog as they are the
precursors of smog and ozone." Due to the hazardous e ect of
VOCs on human health and the environment, the inves-
tigations to remove these harmful VOCs from the environment
have been accelerated.*>* Conventional technologies include
absorption, catalytic oxidation, membrane separation, and
biological treatment, which are not economical for the
elimination of dilute VOCs from a large volume of gas

Recently, nonthermal plasma (NTP) has drawn interest |n
VOCs abatement due to its high reactivity at mild conditions.”

However, the low product selectivity and the generation of
undesired and toxic bgproducts limit the industrial application
of NTP technology.™ It has been already demonstrated that
the product selectivity of NTP technology can be improved by
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combining it with catalysis. The heterogeneous catalg/st could
be put inside or after the plasma discharge region.” For the
former, the catalyst could utilize the short-lived species formed
in the discharge process and enhance the electric eld
intensity.” However, for the latter, the catalyst could utilize
the ozone and remove byproducts generated in the discharge
region.” In addition, numerous parameters such as gas ow
rate, discharge power, and water can in uence the reaction
performance of the NTP-catalytic system via changing the
reaction time and chemical species formed in the discharge
process and catalytic reaction process.®’

Catalysts occupy a vital position in the NTP-catalytic system
by e ectively utilizing the reactive oxygen species produced in
the discharge zone, thus leading to a synergetic e ect between
NTP and catalyst.° So far, numerous works have been carried
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out for the mvestlgatlon of the metal-based catalysts in NTP-
catalytic systems.”*~ Among the various metal-based catalysts,
transition metal catalysts are superior to supported noble metal
catalysts owing to their low cost. In particular, manganese and
cobalt-based catalysts have been extensively investigated based
on their remarkable potential applications in decomposing
ozone at room temperature.** For Mn-based catalysts, the high
catalytic activity is attributed to their multiple oxidation states
and high activity and Co-based catalz/sts are reported to show
e cient total oxidation of VOCs.** Also, the mixed oxide
catalysts possess better catalytic activity than the single oxide
catalysts due to the synergistic e ect between di erent metal
oxides. Besides the loaded metals/metal oxides, the catalyst
supports also occupy a major role in the e ectiveness of the
catalysts.”* HZSM-5 zeolite has been widely used for VOCs
elimination due to its relatively high adsorption ability, special
mesoporous structure, and high hydrothermal stability.™* Thus,
Mn and Co oxides loaded on the HZSM-5 may enhance the
catalytic properties of the composite materials. However, up
until now, very few works have been done on a thorough and
systematic study of VOCs degradation over mixed oxides
loaded on the HZSM 5, especially in the post NTP-catalytic
(PPC) process.***

In this work, acetaldehyde was used as a model VOCs due to
its wide usage and its hazardous e ect on human health and
the environment. A PPC system combining a DBD reactor and
various Mn Co/HZSM-5 catalysts were used to study the
acetaldehyde decomposition. The e ect of di erent Mn Co/
HZSM-5 catalysts and various experimental parameters such as
gas ow rate and SED on the removal e ciency of
acetaldehyde, CO, yield, CO, yield, and energy yield (EY)
were investigated. To obtain deeper insight into the in uence
of the catalyst properties on acetaldehyde removal, multiple
characterization techniques like X-ray di raction (XRD), high-
resolution-transmission electron microscopy (HR-TEM), and
hydrogen-temperature-programmed reduction (H,-TPR) were
performed. In addition, the reaction Kkinetics for the
degradation of acetaldehyde in the PPC system over di erent
catalysts were discussed using a steady-state approximation
method based on the experimental results.

2. EXPERIMENTAL SECTION

2.1. Catalyst Synthesis and Characterization. A
deposition-precipitation method was used to synthesize the
Mn Co/HZSM-5 (the molar ratio of Mn/Co = 1) catalysts.
Mn(NO,),-4H,0 and Co(NO3);-9H,0 were selected as the
precursors, meanwhile, the HZSM-5 zeolite was used as the
support. In a typical synthesis, in the rst step aqueous
solutions of Co(NO5), (0.5 mol L *) and Mn(NO3;), (0.6 mol
L 1) were mixed by continuous magnetic stirring. Sub-
sequently, HZSM-5 zeolite was added to the solution under
stirring. Then, KMnO, (0.4 mol L ') was introduced into the
acquired slurry and stirred at 60 °C for 1 h. After that, the pH
of the above mixture was tuned to 9 by dropwise addition of
Na,COj; and the stirring at 60 °C was continued for 4 h. Lastly,
the sample was obtained by Itration, drying, and calcination at
500 °C for 3 h. Also, MnO, and Co;0, loaded on the HZSM-5
were synthesized using the above-mentioned method. The
samples were expressed as Mn/HZ-5, Co/HZ-5, and MnCo/
HZ-5.

The characterizations of the catalyst samples were analyzed
by XRD, Raman, the N, adsorption desorption isotherms,
inductively coupled plasma emission spectrometry (ICPES),
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scanning electron mlcroscopy (SEM) HR-TEM, XPS, and H2
TPR as described in our previous studies.*®*’ The ammonia-
temperature programmed desorption (NH3-TPD) was oper-
ated to study the surface acidity of the catalyst. The NH;-TPD
analysis started with the pretreatment of 200 mg of catalyst in
argon at 300 °C for 1 h. Then, the adsorption of NH; (10%, Ar
as balance gas) was conducted at 30 °C until saturation which
took about 1.5 h. Subsequently, the excess NH; was wiped
away by ushing the sample with pure argon. Finally, the NH;
desorption was operated by heating the sample to 800 °C.
2.2. Experimental Setup. As displayed in Figure 1, the
overall setup included a gas supply system, PPC system, and

AC high voltage power supply] OSCileﬁD
—MFC =

DBD reactor

+{C—-5]/ Ozone monitor

| L T - catalyst ?f ?m}: IZ[:{’?W]/
Mlxm;, chamber ﬁ m FTIR
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Figure 1.Diagram of the experimental setup.

analysis system. The detailed description of the DBD reactor,
oscilloscope, ac high-voltage power supply, as well as the
calculation of the discharge power were reported in our
previous work.*®

2.3. Post NTP Catalysis. The post NTP catalysis for
acetaldehyde removal was conducted at ambient pressure and
temperature using dry air with a xed volume ratio of O,/N,
(1/4) as carrier gas. The desired initial concentration of
acetaldehyde was obtained by tuning the ratio of purge of
acetaldehyde and carrier gas. The gas ow rate was tuned by a
mass ow controller (MFC, Sevenstar DO7-series, China). The
initial concentration of acetaldehyde was set as 200 ppm. The
range of the total gas ow rate was 0.5 1.5 L min 1. Catalyst
(0.2 g) was put in the downstream of the NTP discharge zone
by quartz-wool loading. It was puri ed at 300 °C for 3 h in the
N, ow before use. The concentration of acetaldehyde was
measured continuously using an online Fourier transform
infrared spectroscopy (Gasmet DX4015, Finland) with a xed
resolution of 8 cm 1. The concentration of CO, and CO were
determined by gas chromatography (GC 9790 Il, Fuli, China).
A UV-based ozone (Oj) analyzer (2B, 106M) was used to
analyze the concentration of O,. All the reported experimental
data in this work are an average of results obtained in three
repeated experiments.

The acetaldehyde removal e ciency ( geetaigenyce): SED, EY,
CO, yield, and CO, yield were de ned as follows:

)= [acetaldehyllg; S [ acetaldehyge
0 -—
acetaldehycg! [acetaldehyﬁget
x 100% 1)
- i h
D (3 8Y)= discharge power (W )( 60 (s mu?l)
gas flow rate (L mih )
2
EY (g KWL )= Macetaldenyd® acetaldehyde, 3.6
SED Vi 3)
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CO, yield (%)

_ [cOJ
2 x ([acetaldehybg:S [ acetaldehyds )
x 100% 4
CQ, yield (%)
- [CO +[CO
2 x (lacetaldehyPg,S [ acetaldehyge )
x 100% (5)

where Micetaigenyde aNd Vi, represents the molar mass of
acetaldehyde (g mol ) and molar volume (L mol 1).

3. RESULTS AND DISCUSSION

3.1. Characterizations of Catalysts. 3.1.1. Crystal
Structure and Surface CharacteristiéRD was carried out
to determine the crystalline structure of the fresh catalysts. As
displayed in Figure 2a, all the XRD patterns showed a series of
peaks which are characteristics of HZSM-5,® revealing that the
crystalline texture of HZSM-5 remained after Mn and Co

Figure 2XRD (a) and Raman (b) patterns of the samples: a, HZSM-
5; b, Mn/HZ-5; ¢, Co/HZ-5; d, MnCo/HZ-5.
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loading. However, with Mn and Co loading, the peak intensity
of the characteristics peaks of HZSM-5 weakened, while no
obvious new peak appeared. This phenomenon is in reasonable
agreement with the literature,”**® which reveals a good
dispersion of manganese and cobalt oxides on the HZSM-5
surface. The intensity of XRD peaks of MnCo/HZ-5 was lower
and wider compared with those of Mn/HZ-5 and Co/HZ-5,
indicating the possible interaction between Mn and Co species
on the catalysts, which improves the dispersion and inhibits the
crystallization. This also indicated that the bimetal oxides
possess a relative smaller crystalline size in comparison with
the monomer oxides. As expected, the particle size (Table 1,

Table 1. Characterization of the Fresh Catalysts

atomic mass
ratio®

A?Jrfacs:-l DPore Vgore 1 b
sampe.  Mn  Co  (m°g) (M) (M g ) Sk
HZSM-5 2.06 021 2.06 48.02
Mn/HZ-5 985 0 4.22 0.27 422 4941
Co/HZ-5 0 9.92 3.56 0.25 3.56 47.16
MnCo/ 988 996 4.72 0.30 4.72 43.19

HZ-5
3From ICPES. PParticle size, which was calculated based on the
Scherrer equation.

calculated using the Scherrer equation®®) of MnCo/HZ-5 was
lower than Mn/HZ-5 and Co/HZ-5. The smaller particle size
of the catalyst provides more active sites which accelerate the
chemical reactions.

Raman spectroscopy was performed to further study the
crystallographic structure of the fresh catalysts. As presented in
Figure 2b, no obvious band was found for the HZSM-5
catalysts. The characteristic bands for the Mn/HZ-5, Co/HZ-
5, and MnCo/HZ-5 were in the range of 400 800 cm 1. For
the Mn/HZ-5 catalyst, the bands between 500 700 cm * were
related to the stretching modes of MnOg octahedra, and the
band at 384 cm * was related to the existence of -MnO,.** As
for Co/HZ-5 catalyst, the bands centered at 470 (E,), 677
(Agp), 605 (Fy), 511 (Fy), and 189 (F) cm ' were the
typical bands of spinel Co;0,'* The bands of MnCo/HZ-5
were broader and the peak intensities were lower in
comparison with Mn/HZ-5 and Co/HZ-5, indicating the
interaction of Co and Mn species,*> which was in line with the
result of XRD.

Table 1 presented the pore diameter, speci ¢ surface area,
and total pore volume of the fresh catalysts. It is clearly shown
that the HZSM-5 possesses the largest speci ¢ surface area
(410.50 m? g ). Meanwhile, the MnCo/HZ-5 has the highest
pore diameter (4.72 nm) and total pore volume (0.30 cm?
g 1). However, the average surface area decreased after the Mn
and Co species loaded on the HZSM-5, and this was related to
the blockage of pores by the presence of Mn and Co
species.**'% In addition, the atomic mass ratios of Mn and Co
of the di erent catalysts obtained by ICP analysis were
reasonably close to the theoretical values within experimental
and measurement error (Table 1).

3.1.2. Morphology and Elemental Analy3ise morphol-
ogy and surface texture of the prepared catalysts were recorded
by SEM, TEM, and HR-TEM. It could be found that the
surface of HZSM-5 is very smooth, and some granules
appeared after the Mn and Co loading on HZSM-5 as
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presented in Figure S1. As seen in HR-TEM images (Figure 3),
the interplanar distances of 0.346 and 0.271 nm in the Mn/

Figure 3.TEM and HRTEM patterns of Mn/HZ-5 (a,b), Co/HZ-5
(c,d), and MnCo/HZ-5 (g/f).

HZ-5 catalyst were attributed to the (220) plane of MnO,
(JCPDS 44-141) and (222) facet of Mn,O5 (JCPDS 41-1442),
respectively. The lattice fringes of the Co/HZ-5 catalyst with a
spacing of 0.285 nm could be attributed to the (220) plane of
Co03;0, (JCPDS 74-1657). The interplanar spacing of 0.489,
0.271, and 0.285 nm was observed in the MnCo/HZ-5
catalyst, which corresponds to the (200) plane of MnO,, (222)
plane of Mn,O3, and (220) plane of Co;0,, respectively. The
EDX mapping of MnCo/HZ-5 catalysts (Figure S2) exhibits
high and uniform dispersion of Mn and Co species.

The super cial elemental composition and their related
oxidation state of the catalysts were analyzed via XPS. Figure
S3 presented the survey spectra of all the fresh catalysts and
used MnCo/HZ-5. As presented in Figure 4, the spectra of Mn
2p displayed two dominant peaks with binding energies of
642.3 and 653.8 eV, which were related to Mn 2p;/, and Mn
2Dy, respectively.”® The Mn 2ps, spectrum was resolved into
three peaks at 643.9, 641.1, and 642.4 eV, relating to Mn*",
Mn?*, ‘and Mn**, respectively.”>** The calculation of the ratio
of (Mn** + Mn*)/Mn was obtained by the quantitative
analysis of Mn 2p,/, spectra. As presented in Table 2, the
(Mn®* + Mn*)/Mn ratio of MnCo/HZ-5 is higher than that
of Mn/HZ-5. The cobalt species were characterized by the Co
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Figure 4.XPS patterns of Mn 2p, Co 2p, and O 1s (a, Mn/HZ-5; b,
Co/HZ-5; ¢, MnCo/HZ-5; d, MnCo/HZ-5 used).

Table 2. Elemental Compositions of M@o/HZSM-5
Catalysts

catalyst Co?*/Co*  Mn*/Mn  Mn*/Mn  0,/0,
Mn/HZ-5 0.48 0.24 34
Co/HZ-5 0.93 224
MnCo/HZ-5 1.07 0.51 031 3.70
MnCo/HZ-5 used 1.01 0.39 0.28 0.32

DOI:10.1021/acs.iecr.9b02668
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2p1/, and Co 2ps, peaks located between 795 and 780.0 eV.*
The XPS spectra of Co 2p;,, was tted with two peaks at 783.2
and 780.8 eV, which was corresponding to Co?* and Co*,
respectively.”® Meanwhile, the small shoulder peak at 788.4 eV
could be ascribed to the shake up satellite peak (Co?*) of
C050,.%° Quantitatively, it was found that the MnCo/HZ-5
catalyst possesses the highest ratio of Co**/Co®. The O 1s
XPS peaks of the as-prepared catalysts were split into two
peaks located at 530.2 + 0.2 eV and 5323 £ 0.2 eV,
corresponding to lattice oxygen seecies (O) and adsorbed
oxygen species (O,), respectively.”” The ratio of O,/O, was
calculated (shown in Table 2), and it decreased in the
following order: MnCo/HZ-5 > Mn/HZ-5 > Co/HZ-5. It was
generally recognized that the O, possess higher activity than
the O,.' Additionally, they occupy a main role in the oxidation
reactions because of their higher mobility. Thus, higher
content of O, is conducive for the production of reactive
oxygen species, which could further accelerate the oxidation of
acetaldehyde.

3.1.3. BTPR and NHTPD AnalysisThe reduction
characteristics of the fresh catalysts were determined by H,-
TPR. As shown in Figure 5a, there was no distinct H,

Figure 5.H,-TPR (a) and NH5;-TPD (b) patterns of catalysts (a,
HZSM-5; b, Mn/HZ-5; ¢, Co/HZ-5; d, MnCo/HZ-5)

consumption peaks for the HZSM-5 support within the
measured temperature range. For Mn/HZ-5, two obvious
reduction peaks located around 343 and 460 °C were related
to the reduction process of manganese from Mn** to Mn** and
then to the reduction from Mn** to Mn?**® The Co/HZ-5
catalyst has a broad peak at 322 °C, which was in line with the
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reduction process of Co®*" to Co* and Co®* to Co.*** A
higher temperature at 712 °C could be in association with the
strong interaction between the support and Co species.****
For MnCo/HZ-5 catalyst, the overlapped reduction peak at
266 °C was assigned to the reduction of Mn**  Mn®* and
Co* Co?, and the overlapped peak at 379 °C was
attributed to the reduction of Mn**  Mn?*" and Co?*  Co.
A higher reduction temperature at 656 °C was related to the
strong interaction between Mn, Co species with support.®***
In addition, it can be concluded from the lower temperature
shift and higher peak area of the reduction peak that the redox
potential of MnCo/HZ-5 catalyst was higher than that of Co/
HZ-5 and Mn/HZ-5. This could be related to the interaction
between cobalt and manganese species in MnCo/HZ-5
catalyst.

As presented in Figure 5b, two desorption peaks were found
in the NH5-TPD curves at 50 350 °C and 350 600 °C. The
former was ascribed to the successive desorption of NH; from
Lewis acid sites, while the latter was ascribed to the desorption
of NH, from Bronsted acid sites.** Accounting to the total
peak area between 50 and 600 °C, the amount of the total
acidity decreased in the following sequence: HZSM-5 > Mn/
HZ-5 > MnCo/HZ-5 > Co/HZ-5. The loading of Mn and Co
species on the HZSM-5 decreased the amount of acidity,
which was in agreement with previous studies.***> Moreover,
this phenomenon also indicated that the acidity of these
catalysts are not a determining factor for the catalytic activity
because of the fact that the most active catalyst (MnCo/HZ-5)
showed a lower acidity.

As presented in section 3.1, all the Mn Co/HZ-5 catalysts
showed no obvious changes in textural properties when
compared to HZSM-5 support. The XPS results showed that
the MnCo/HZ-5 catalyst possessed the highest (Mn®*" +
Mn*")/Mn, Co?*/Co*, and O,/O, ratios. The redox pairs such
as Mn** O? Co? could promote the mobility and content
of reactive oxygen species favor the oxidation of gas
pollutants.* The H,-TPR pro les showed that the MnCo/
HZ-5 catalyst possessed the highest reducibility, indicating its
easier activation and higher mobility of surface oxygen
species.” All the above-mentioned properties of MnCo/HZ-5
catalysts are in close relation to its catalytic activity and
consequently in uenced the reaction performance.

3.2. Acetaldehyde Removal in the PPC System.
3.2.1. Eect of SEDFigure 6 presented the in uence of
SED on the cetaigenyqer EY, CO, yield, and CO, yield in the
NTP alone and PPC system. The gas ow rate was set as 1 L
min L It is obvious that increasing the SED promoted the

acetaldenyger CO> Yyield, and CO, yield for all the cases. However,
the EY varied inversely with the SED. Taking the NTP alone
system for instance, increasing the SED from 80.25 to 380.03 J
L % the acetaidenydes CO; Yield, and CO, yield took an increase
from 32.41% to 77.14%, 8.70% to 23.70%, and 15.33% to
39.50%, respectively, while the EY yield decreased from 5.23 to
2.63 g kWh L It is known that the higher energy favors the
generation of high energy electrons and active species,” which
are conducive for the formation of various oxidative species
produced via collisions between carrier gas molecules and
high-energy electrons, further accelerating the acetaldehyde
oxidation process. It was reported that acetaldehyde dissoci-
ation in the NTP discharge region is divided into three
pathways involving high-energy electrons, ions, and active
radicals.® However, other report claimed that the reactions
between VOCs and ions are negligible.*® Klett et al.*” also

DOI:10.1021/acs.iecr.9b02668
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Figure 6 E ect of SED and catalysts on (a) the acetaldehyde removal
e ciency, (b) EY, (c) CO, yield, and (d) CO, yield.

reported that acetaldehyde destruction is controlled by two
body reactions: high-energy electron impact and active radical
(such as atomic oxygen, excited and metastable nitrogen
species, hydroxyl radicals). Jia et al.” reported that reactions
between acetaldehyde molecules and high-energy electrons
and N,* played a dominant role in the acetaldehyde
destructions and the relative contribution of reactions between
acetaldehyde and reactive species (N,*, O, OH, and O;) and
high-energy electrons are dependent on the SED. The impact
of the high-energy electrons on the carrier gas produces
reactive species (such as O, N,(A), N,*). The generation of
these active species allows the stepwise decomposition and
oxidation of acetaldehyde into CO,, intermediates and other
byproducts via the following equations: eqs 6 18,137

CH,CHO+e CH CHO (6)
CH,CHO+ N HCN Hs CHO )
CHCHO + Ny(A) CHy C® NyX) (8)
CHCHO + Ny(A) CHy CH® N{X) )
CH,CHO+ N, CHLCO® H NyX) (10)
CH,CHO + N, CHy CH® NyX) (11)
CH,CHO+ O OH CH@ (12)
CH,CHO+ O C® CH OH (13)
CH,CHO+ OH H,@ CH;00 (14)
CH,CHO+ OH H,@ CH,CHO (15)
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CHCO+ (0/0) CO% CHQ (16)
CH;+0 C® M H, a7
CHCO+ 0, CG CHO, (18)

3.2.2. Eect of Di erent CatalystsAs shown in Figure 6, in
comparison with the NTP alone system, the cetaigenyger EY
yield, CO, yield, and CO, yield were enhanced in the PPC
system irrespective of the catalyst used. The presence of
HZSM-5 catalyst markedly promoted the acetaldehyde
removal e ciency. The large speci ¢ surface area of HZSM-
5 provides many adsorption sites which prolong the detention
time of the pollutant in the reaction zone, leading to higher EY
and  sceraidenyge: HOWever, the CO, yield and CO, yield were
still too low, and it is essential to add active species to enhance
the CO, and CO, yield. The addition of Mn and Co oxides
markedly improved the CO, and CO, yield, with MnCo/HZ-5
exhibiting the highest catalytic activity. This increased catalytic
performance of MnCo/HZ-5 was associated with the
dispersion of Co and Mn species on the HZSM-5 support.
As already presented in the H,-TPR results, the MnCo/HZ-5
catalyst possesses a higher intensity of reduction peaks and
lower reduction temperature. This manifested that MnCo/HZ-
5 catalyst owned the highest mobility of surface oxygen species,
promoting the acetaldehyde oxidation process. The XPS result
also showed that MnCo/HZ-5 catalyst possess a higher
quantity of adsorbed oxygen species, which could be conducive
to the oxidation of acetaldehyde. Hence, more production of
active surface oxygen species could boost the total oxidation of
acetaldehyde.

After treated in the discharge region, the unreacted or partly
reacted gas mixture was subsequently adsorbed onto the
catalyst via surface active sites, which could further react with
active surface oxygen to produce CO, and H,0.° The
interaction between cobalt and manganese species promotes
the mobility of oxygen on the catalyst surface and enhances the
conversion of lattice oxygen and oxygen molecules captured
from the carrier gas to active oxygen species. In addition, ozone
was dissociated to atomic oxygen by catalysts (as discussed
below). Thus, active surface oxygen species had a crucial
position in the catalytic reactions, which was generated from
the lattice oxygen and ozone formed in the discharge region.*

3.2.3. Inuence of Gas Flow Rat&he e ect of gas ow
rate on the acetaldehyde degradation in the NTP alone and
PPC system is presented in Figure 7. As expected, the augment
of gas ow rate decreased the CO, yield, CO, yield and

acetaldenyge: FOr €xample, in case of NTP combined with
MnCo/HZ-5 catalyst, as the total gas ow rate increased from
05 L min ! to 1.5 L min % the CO, yield, CO, yield, and

acetaldenyde  d€Creased from 36.64% to 18.07%, 50.20% to
32.85%, and 99.27% to 75.24%, respectively. The retention
time of acetaldehyde in the plasma discharge region was
decreased due to the augment of the gas ow rate, which leads
to insu cient reactions between the gaseous pollutants,
reactive species, and high-energy electrons.>“° This results in
decreased CO, yield, CO, yield, and  ;ceqaigenyqe- HOWever, the
augment of the gas ow rate led to a prominent improvement
in the energy yield. The energy yield increased from 2.70 to
6.15 9 kWh * when the gas ow rate raised from 0.5 to 1.5 L
min -,

3.3. O; Formation. The concentration of O3 generated in
the NTP alone and PPC system was shown in Figure 8. In all
the cases, the amount of Oz produced increased with the
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Figure 7.In uence of gas ow rate on () the acetaldehyde removal
e ciency, (b) EY, (c) CO, yield, and (d) CO, yield.

Figure 8.In uence of di erent catalysts on the production of Oj in
the NTP and PPC systems.

increase of SED. The PPC system packed with Mn  CoO,/HZ
produced less O5 in comparison with the NTP system. For
instance, the concentration of O; at the SED of 200.1 JL *in
NTP system and PPC system with MnCo/HZ-5 was 733.60
and 226.33 ppm, respectively. The generation of O5 was via the
collisions between high-energy electrons and oxygen in the
discharge region according to egs 19 21.*° The O, thus
formed was decomposed to form active atomic oxygen by the
catalyst as shown in eqs 23 and 24 (* donated as surface active
sites of the catalyst),® which participates in the pollutant
oxidation reactions.

€+0 & O (19)
e+0 © O ¢€ (20)
0+0, G (21)

O;+* O O, (22)
O+ 0O O 0O (23)
O Oy (24)

3.4. Reaction Kinetics. In the PPC system, acetaldehyde
molecules were rst destructed by high-energy electrons, ion
collisions, and reactive radicals like OH and O and then
oxidized by surface active oxygen species.'®*° The rate
coe cient of each elementary reaction included in the NTP-
catalytic degradation process of VOCs is di cult to determine
as a result of that the concentrations of radicals and
intermediate species during the destruction were di cult to
detect. Moreover, no comprehensive model has been
established to describe the interactions between NTP and
catalysts for the reactions in the whole NTP-catalytic
degradation process for VOC elimination. Thus, we selected
a simple model to describe the PPC system, assuming a plug

OW.40’41

The rate of acetaldehyde degradation ( cpscho) could be

expressed as

g%: (kex Gt kionlCiot KvGt & Kk &

+ KyCGi+ oy Gr) X Gucro (25)
where G, Gon, Cni Co, Coz, Gy Con, and Cenacrio represent

the concentration of high-energy electrons, ions, N-containing
species, O , H , O3, OH , and acetaldehyde, and k, Kion, Kni: Ko,
Kos, ku, and ko represent their respective reaction rate
constants.

Table 3 showed the reaction constants of the main reactions
in the acetaldehyde degradation process. The e ect of O3 on

Table 3. Main Reactions Cited in the Paper

K300
(cm®mol s ) ref

reaction

CH3;CHO +e products + e 4

CH;CHO + N  HCN + H, + CHO 2 %10 * 43
CH,CHO + N, (A)  CHg+ CHO + N, (X) 1x10% 37
CH CHO + N¥  CHj; + CHO + Ny(X) 5x 10 10 37
CH,CHO+ O OH + CH,CO 46 x 10 37
CH,CHO+0O,  products 34x10% 44
CH,CHO +H  H, + CH,CO 7.95 % 10 ¥ 45
CH,CHO + OH  H,0 + CH,CO 15x10 ¢ 37

acetaldehyde elimination was ignored, owing to its small
reaction constant. The rate constants of the reaction between
N, H, and acetaldehyde were lower than that of N, metastable
states. Thus, the e ect of N and H radicals on acetaldehyde
abatement was ignored. The rate constant of the reaction
between O species and acetaldehyde were lower than that of
N, metastable states, but the concentration of O is higher than
N, metastable states as reported in the previous report.*’
Meanwhile, the e ect of OH was also neglected due to its low
density.*> The electron collisions were neglected due to their
short lifetime and low density. In the view of that the
concentrations of N, metastable states and O species are
constant under xed air pressure and electric eld** eq 25
could be simpli ed as

= dc
SE = kx Cacetaldehyﬁd (26)
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where Kk is donated as the apparent rate constant.
Integrating eq 26 resulted in the following:

&in Sout = ¢
in (27)
In view of eq 1, eq 27 could be expressed as follows:
éh’l(l S acetaldehyt}e: kt (28)
Figure 9 showed the relationship between the In(1

acetaldehyde) @nd time for di erent catalysts. The slope of the

Figure 9.Rate constants of acetaldehyde removal in the NTP and
PPC systems.

Table 4. Apparent Rate Constants and Regression
Coe cients in the NTP and PPC Systems

NTP HZSM-5  Mn/HZ-5 Co/HZ-5 MnCo/HZ-5
k 0.17 0.25 0.34 0.36 0.44
R 0.9967 0.9974 0.9943 0.9945 0.9958

line gives the apparent rate constant. As shown in Table 4, all
the regression coe cients (R?) were greater than 0.99,
indicating that the linear tting was statistically acceptable.
Compared to the NTP alone system, the introduction of
catalysts increased the rate constant, with the MnCo/HZ-5
catalyst exhibiting the highest rate constant. This was in
agreement with the highest acetaldehyde removal e ciency
obtained by the combination of NTP and MnCo/HZ-5
catalyst. This phenomenon indicated that the combination of
NTP with the catalysts enhanced acetaldehyde oxidation.

4. CONCLUSIONS

The Mn Co/HZSM-5 catalysts synthesized via the deposi-
tion precipitation method were applied in the PPC system for
acetaldehyde removal. The combination of the catalyst with
NTP showed a notable advance in the acetaldehyde removal
e ciency, CO, yield, CO, yield, and EY and greatly inhibited
the formation of O;. The mixed MnCo/HZSM-5 showed the
best activity in the PPC system for acetaldehyde removal,
which was related to the crystal structure, redox property, and
surface adsorbed oxygen species. Additionally, the acetalde-
hyde removal e ciency, CO, yield, and CO, yield increased
with increasing the SED but decreased with increasing the gas

ow rate. However, the EY was reduced with the increase of
SED and promoted with the augment of the gas ow rate.

14726

Moreover, the reaction Kinetics results showed that the
acetaldehyde removal in the PPC system was a typical rst-
order reaction. The addition of catalysts markedly promoted
the acetaldehyde degradation process, with the mixed MnCo/
HZ-5 exhibiting the highest apparent rate constant.
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ABBREVIATIONS

A iace = SPeCi € surface area

Cchscho = concentration of acetaldehyde

C. = concentration of high-energy electrons

Cy, = concentration of hydrogen

Cion = concentration of ions

Cy = concentration of nitrogen-containing species
Co = concentration of active oxygen

CO, = carbon dioxide

Cos = concentration of ozone

CO = carbon monoxide

Con = concentration of hydroxide radical

CO, = carbon monoxide + carbon dioxide
Dpore = average pore diameter

DBD = dielectric barrier discharge

EDX = energy dispersive X-ray

eq = equation

EY = energy yied

FTIR = Fourier transform infrared spectroscopy
GC = gas chromatography

H,-TPR = hydrogen temperature-programmed reduction
HZ-5 = HZSM-5

H,O = water
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HR-TEM = high-resolution-transmission electron micros-
copy

ICPES = inductively coupled plasma emission spectrometry
k. = reaction rate constant of high-energy electrons

ky = reaction rate constant of hydrogen

kion = reaction rate constant of ions

ky = reaction rate constant of nitrogen-containing species
ko = reaction rate constant of active oxygen

kos = reaction rate constant of ozone

ko = reaction rate constant of hydroxide radical
Micetaigenyde = Molar mass of acetaldehyde

MFC = mass ow controller

NH3-TPD = ammonia-temperature programmed desorption
NTP = nonthermal plasma

N, = nitrogen
O3 = ozone
O, = oxygen

ppm = parts per million
Sariicle = average pore diameter
SED = speci ¢ energy density
SEM = scanning electron microscopy
VOCs = volatile organic compound
V,,, = molar volume
Voore = total pore volume
XPS = X-ray photoelectron spectroscopy
XRD = X-ray di raction
acetaldehyde = acetaldehyde removal e ciency
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